# Patterns of Ancestry, Signatures of Natural Selection, and Genetic Association with Stature in Western African Pygmies

Joseph P. Jarvis<sup>1</sup>, Laura B. Scheinfeldt<sup>1</sup><sup>9</sup>, Sameer Soi<sup>1</sup><sup>9</sup>, Charla Lambert<sup>1</sup><sup>9</sup>, Larsson Omberg<sup>2</sup><sup>9</sup>, Bart Ferwerda<sup>1</sup>, Alain Froment<sup>3</sup>, Jean-Marie Bodo<sup>4</sup>, William Beggs<sup>1</sup>, Gabriel Hoffman<sup>2</sup>, Jason Mezey<sup>2,5</sup>, Sarah A. Tishkoff<sup>1,6</sup>\*

1 Department of Genetics, University of Pennsylvania, Philadelphia, Pennsylvania, United States of America, 2 Department of Biological Statistics and Computational Biology, Cornell University, Ithaca, New York, United States of America, 3 UMR 208, IRD-MNHN, Musée de l'Homme, Paris, France, 4 Ministere de la Recherche Scientifique et de l'Innovation, Yaounde, Cameroon, 5 Department of Genetic Medicine, Weill Cornell Medical College, New York, New York, United States of America, 6 Department of Biology, University of Pennsylvania, Philadelphia, Pennsylvania, United States of America

# Abstract

African Pygmy groups show a distinctive pattern of phenotypic variation, including short stature, which is thought to reflect past adaptation to a tropical environment. Here, we analyze Illumina 1M SNP array data in three Western Pygmy populations from Cameroon and three neighboring Bantu-speaking agricultural populations with whom they have admixed. We infer genome-wide ancestry, scan for signals of positive selection, and perform targeted genetic association with measured height variation. We identify multiple regions throughout the genome that may have played a role in adaptive evolution, many of which contain loci with roles in growth hormone, insulin, and insulin-like growth factor signaling pathways, as well as immunity and neuroendocrine signaling involved in reproduction and metabolism. The most striking results are found on chromosome 3, which harbors a cluster of selection and association signals between approximately 45 and 60 Mb. This region also includes the positional candidate genes *DOCK3*, which is known to be associated with height variation in Europeans, and *CISH*, a negative regulator of cytokine signaling known to inhibit growth hormone-stimulated *STAT5* signaling. Finally, pathway analysis for genes near the strongest signals of association with height indicates enrichment for loci involved in insulin and insulin-like growth factor signaling.

Citation: Jarvis JP, Scheinfeldt LB, Soi S, Lambert C, Omberg L, et al. (2012) Patterns of Ancestry, Signatures of Natural Selection, and Genetic Association with Stature in Western African Pygmies. PLoS Genet 8(4): e1002641. doi:10.1371/journal.pgen.1002641

Editor: Joshua M. Akey, University of Washington, United States of America

Received November 16, 2011; Accepted February 21, 2012; Published April 26, 2012

**Copyright:** © 2012 Jarvis et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Funding:** This work was funded by NSF grants BCS-0196183 and BCS-0827436 and by NIH grants DP1-OD-006445-01, R01-GM076637, and R01-GM083606 to SAT. JPJ is supported by an NIH, NRSA fellowship (1F32HG005290-01). BF is supported by a Rubicon grant of The Netherlands Organization for Scientific Research. LO and JM are funded by the Department of Genetic Medicine (Weill Cornell Medical College), NIH grant 1P50MH079513-02, and NSF grants DEB-0922432 and IOS-1026555. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

\* E-mail: tishkoff@mail.med.upenn.edu

• These authors contributed equally to this work.

# Introduction

The term 'Pygmy' is applied to human populations whose adult males exhibit an average height of ~150 cm or less, although thresholds between 140 and 160 cm have been employed [1]. Such groups are found all over the world including Africa, Asia, and the Americas, tend to live in tropical environments, have high levels of pathogen exposure, and practice a predominantly hunting and gathering lifestyle [2]. Studies of African mtDNA diversity suggest a time to most recent common ancestor (TMRCA) between African Pygmies and West African Bantu groups of  $\sim 60-$ 70 thousand years ago (kya) and between Eastern (Efe, Mbuti, Batwa, Babinga, etc.) and Western (Biaka, Baka, Bakola, Bedzan, etc.) Pygmy groups ~10 to 27 kya [3,4,5,6,7]. Approximately 4-5 kya, Bantu speaking populations practicing slash-and-burn agriculture expanded into the forest habitats populated by the ancestors of modern Pygmy populations [8]. This secondary contact resulted in admixture, predominantly involving maternal Pygmy and paternal Bantu ancestors that remains on-going [6], as well as a cultural exchange that eliminated the Pygmy's ancestral language [9].

Short stature in human Pygmy populations is thought to be adaptive and its global distribution indicates either ancient common ancestry among tropical hunter-gatherers or extensive convergent evolution [2]. Several selective mechanisms have been proposed to explain a fitness advantage for small body size in tropical, forest-dwelling, hunter-gatherer populations. These include resistance to heat stress under humid forest conditions [2], reduced caloric requirements in a relatively food-limited environment [2], and a life-history trade-off between cessation of growth and early reproduction [1,10]. However, the timing of the putative selective episodes and emergence of contemporary Pygmy morphology and physiology remain unclear. It is also unclear how the genetic architecture of body size in these groups may differ from that observed in other West African populations [11] and Europeans, in which  $\sim 180$  loci explaining  $\sim 10\%$  of the phenotypic variance have been identified [12].

## **Author Summary**

Africa is thought to be the location of origin of modern humans within the past 200,000 years and the source of our dispersion across the globe within the past 100,000 years. Africa is also a region of extreme environmental, cultural, linguistic, and phenotypic diversity, and human populations living there show the highest levels of genetic diversity in the world. Yet little is known about the genetic basis of the observed phenotypic variation in Africa or how local adaptation and demography have influenced these patterns in the recent past. Here, we analyze a set of admixing Bantu-speaking agricultural and Western Pygmy hunter-gatherer populations that show extreme differences in stature; Pygmies are  $\sim$ 17 cm shorter on average than their Bantu neighbors and among the shortest populations globally. Our multifaceted approach identified several genomic regions that may have been targets of natural selection and so may harbor variants underlying the unique anatomy and physiology of Western African Pygmies. One region of chromosome three, in particular, harbors strong signals of natural selection, population differentiation, and association with height. This region also contains a significant association with height in Europeans as well as a candidate gene known to regulate growth hormone signaling.

To date, a variety of physiological traits related to stature have been examined in African Pygmy populations [13,14]. These studies documented normal levels of circulating human growth hormone (HGH) but altered glucose homeostasis, insulin secretion, and free fatty acid profiles following HGH, glucose, and arginine challenges [13,14]. These observations suggest peripheral subresponsiveness to the effects of HGH in various tissues. Subsequent work noted a significant difference in overall growth rate in Pygmies only during puberty and suggested a reduced level of IGF-1, whose production is stimulated by HGH, is more likely responsible for short stature in adults [10,15,16]. In addition, both a reduction in IGF-1 receptor expression and decreased signal transmission in response to physiological concentrations of IGF-1 were observed in immortalized Pygmy T-cells [16,17]. The observations of low levels of high-affinity HGH binding protein in both African and Philippine Pygmies [18,19], and a severe underexpression of HGH receptor (HGHR) in the Bakola Pygmies [20] suggest a deficiency in HGH receptor activity contributes to short stature as well.

With the goal of providing a genomic perspective on the adaptive and genetic basis of short stature in Pygmies, we analyzed genetic and phenotypic data in a set of neighboring Western Pygmy (Baka, Bakola, and Bedzan) and Bantu-speaking (Tikar, Ngumba, Lemande) agricultural populations from Cameroon. Our sample includes 67 Pygmy and 58 Bantu individuals genotyped using the Illumina 1M SNP array. Of these, 57 Pygmy and 39 Bantu individuals have corresponding phenotype data for height. We performed several genome-wide scans for positive selection, including the calculation of FST and locus specific branch length (LSBL) metrics, as well as iHS and XP-EHH analyses. Since regions exhibiting selection signals are likely enriched for functional variants contributing to adaptive differences in phenotype between populations, we summarize the types of loci found in these regions using pathway enrichment analyses. Finally, using both SNP genotypes directly as well as ancestry estimates that facilitate admixture mapping approaches, we performed association tests within candidate regions identified in our selection scans as well as for SNPs near genes in candidate

signaling pathways. This multifaceted approach identified several genomic regions showing strong signals of selection and association, some of which contain candidate genes for height variation. Together, our results provide unique insights into both the complex adaptive process that took place in the ancestors of contemporary Western African Pygmies and Bantu agriculturalists and the genetic architecture of short stature in Western African Pygmies.

## Results

## Population Structure

We observed extensive and significant genetic and phenotypic differentiation (Figure 1, Figure 2, Figure S1) and varying levels of admixture among the Pygmy and Bantu populations. Average levels of Bantu ancestry, as determined by STRUCTURE (K = 2), in the three Western Pygmy populations were 27% (Bakola), 35% (Baka), and 49% (Bedzan) with individual values ranging from 16-73%. Average levels of Pygmy ancestry in the three Bantu populations were <1% (Lemande), 2% (Tikar), and 7% (Ngumba), with individual values ranging from 0-39%. We also observed a highly significant correlation between ancestry and height ( $p = 5.047 \times 10^{-18}$ ) after correcting for the effect of sex (full model  $r^2 = 0.7411$ ,  $r^2$  for sex = 0.4247;  $r^2$  for ancestry = 0.3164). In addition, the effect of ancestry remains significant in a model that also includes Pygmy-Bantu ethnicity as a covariate ( $p = 3.8 \times 10^{-5}$ ). These results are consistent with Becker et al. [21] and indicate a strong genetic influence on height. Similar findings were also observed using Pygmy samples only  $(p_{ancestry} = 0.000216;$  full model  $r^2 = 0.5066$ ;  $r^2 \sec = 0.3744$ ;  $r^2 \operatorname{ancestry} = 0.1322$ ) and the independent set of genome-wide microsatellite markers described in Tishkoff et al. [9] (data not shown).

## Inference of Local Ancestry

The relatively high levels of Bantu ancestry in Western African Pygmy individuals suggests admixture mapping approaches will be useful for identifying loci underlying the short stature phenotype [22]. Thus, we inferred ancestry blocks across haploid Pygmy genomes using the machine learning method SupportMix, which does not require an *a priori* admixture model. Consistent with a model of ancient gene flow, we observe very short average tract lengths of Bantu ancestry (3.1+/-4.6 Mb; Figure 2A). No autosomal regions were significantly enriched for either Pygmy or Bantu ancestry (Figure 2A). However, we do observe regions with reduced levels of switching between ancestry blocks. One such region extends across a 7 Mb region from 46 Mb–53 Mb on chromosome 3 (Figure 2B).

### Scans for Selection: F<sub>ST</sub>/LSBL Analyses

We performed several genome-wide scans for selection that have complementary properties for detecting adaptive events on different timescales and under different selective scenarios [23]. We began by calculating pairwise  $F_{ST}$  values [24] using the Bantu (N = 44) and Pygmy samples (N = 30) estimated to have the lowest levels of admixture in our STRUCTURE analysis. Next, we identified SNPs that are differentiated specifically on the Pygmy lineage using three-way Locus Specific Branch Length analysis (LSBL [25]) comparing Western Pygmy, Bantu, and Hapmap Maasai (N = 45) populations. These cross-population scans for selection (e.g.  $F_{ST}$  and LSBL) are expected to be useful for detecting regionally restricted adaptation, including selection on standing variation [26].

Genome-wide values for pairwise  $F_{ST}$  ranged from 0 to 0.58. High  $F_{ST}$  SNPs are found on all 22 autosomes, but do not appear



**Figure 1. Admixture and population structure.** Plots of principal components A) 1 versus 2 and B) 2 versus 3 calculated using 1M SNP data from 67 Pygmy and 58 Bantu samples. The proportion of variance explained by PCs 1, 2, and 3 is 0.245, 0.0192, and 0.0109, respectively. The three Pygmy ethnic groups are clearly distinguished from their neighboring Bantu populations by the first two PCs, while the third differentiates the three Pygmy groups from one another. C) Regression of height on Pygmy ancestry inferred using STRUCTURE (K=2). Higher levels of Pygmy ancestry are associated with shorter stature. The trend line intersects the intercept from the full model and has a slope equal to the regression coefficient for percent ancestry. D) Visualization of percent ancestry inferred using STRUCTURE at K=2, 3, and 4. doi:10.1371/journal.pgen.1002641.g001

randomly distributed across the genome. Rather, of the 947 SNPs in the top 0.1 percentile of the empirical distribution ( $F_{ST}$ >0.33; Table S1), approximately one third (308) occur in 37 high-density (>2.31/Mb) blocks that average 93 kb in length. Such blocks are defined as a series of four or more high F<sub>ST</sub> SNPs with less than 100 kb separating each and occur on all chromosomes except 11, 13, 16, 20 and 22. Using these criteria, 33% of the most differentiated loci in the genome are found in a total of just 3.42 Mb. The most striking signal (34.7 high  $F_{ST}$  SNPs/Mb) occurs within a 15 Mb region containing 9 separate blocks on chromosome 3 (between 45 and 60 Mb), 7 of which occur between 48 and 53 Mb (Figure 2B, Figure 3, Figure 4A-4E). Interestingly, the highest F<sub>ST</sub> SNP in the region for which there is genotype data from the CEPH human diversity panel (rs7626978 at position 48505831;  $F_{ST} = 0.53$  [27]) shows a striking global distribution in which the minor allele is most common in the Western and Eastern Pygmy, San, and neighboring populations and is nearly absent outside of Africa (Figure S2).

Genome-wide values for Pygmy-specific LSBL SNPs ranged from 0 to 0.42. High Pygmy LSBL SNPs representing the top 0.1 percentile of the empirical distribution (912 with Pygmy LSBL>0.2215) are found on all 22 autosomes (Table S2) and a large proportion of these (306 SNPs, ~34%) are found in 38 physical clusters (visualized as regions of density >2.31 high Pygmy LSBL SNPs/Mb in Figure 3) averaging 99.85 kb in length. Many overlap with high  $F_{\rm ST}$  clusters, as expected. Nine of the 38 Pygmy-specific LSBL regions also occur on chromosome 3 between 45 and 60 Mb.

In order to test the null hypothesis of a random distribution of signals over the genome, we defined non-overlapping 500 SNP bins (average size ~1.77 Mb) and assessed the significance of the number of high  $F_{ST}$  and high Pygmy LSBL SNPs observed in each using the binomial distribution. We identify 82 and 81 bins with over-representation of high  $F_{ST}$  and high Pygmy LSBL SNPs, respectively, that are significant at the p<0.01 level. Among these are three contiguous bins (47560863–49772375 bp, 49773236–52166147 bp, and 52169095–53855083 bp) in the chromosome 3 region showing p-values less than  $1 \times 10^{-12}$  (numbers of high  $F_{ST}$  SNPs = 18, 83, and 11; numbers of high LSBL SNPs = 22, 92, 16). Together, these three bins account for 11.8% and 14.3% of all high  $F_{ST}$  and high Pygmy LSBL markers in the genome.





**Figure 2. Results from local ancestry inference.** A) Genome-wide blocks of Pygmy and Bantu ancestry in Pygmy individuals. B) Blocks of Pygmy and Bantu ancestry in Pygmy individuals for chromosome 3. Shades of blue represent Pygmy ancestry and shades of red admixed Bantu ancestry. Shades are determined by the posterior probability of the estimates; darker colors indicate greater confidence. Each individual is represented by two horizontal rows (one for each haploid genome) and columns represent each window in the genome. Inferred percent Pygmy ancestry for each window is given below. Using SupportMix analysis, we determined the average ancestry block sizes in Pygmies to be 1.7+/-2.4 Mb for Pygmy ancestry. doi:10.1371/journal.pgen.1002641.g002

 $\rm F_{ST}$  and Pygmy-specific LSBL values for X chromosome SNPs are given in Table S3A, S3B). As expected due to its smaller effective population size, these values are higher on average than values calculated for the autosomes. We identify 31 high  $\rm F_{ST}$  and 11 Pygmy specific LSBL SNPs within the top 0.1% of X chromosome values, none of which are near obvious candidate genes for height. Because of the distinct demographic history of the X chromosome, we focus further analyses on only autosomal SNPs.

#### Scans for Selection: XP-EHH and iHS Analyses

Next, we performed iHS [28] and XP-EHH tests of neutrality [29] in the full set of samples to identify regions that may have been targets of recent selective sweeps within and between populations, respectively. Using an XP-EHH threshold of 4.08 (top 0.1%) we identified 400 SNPs, 54 of which fall in the 45–60 Mb region of chromosome 3. The others are spread over the remaining autosomes with the exceptions of chromosomes 15, 18 and 22. Using an even more stringent threshold of 5.0 (the top 0.02% of the distribution) we identified six Pygmy-specific XP-EHH signals, two of which occur on chromosome 3 between 45

and 60 Mb (Table S4; Figure 3, Figure 4). Using iHS we identified 784 SNPs in the Pygmy population and 807 SNPs in the Bantu population showing signals of selection that are widely distributed across the genome (Tables S5, S6). In the Pygmies, 25 of the top iHS SNPs are again found between 45 and 60 Mb on chromosome 3.

# Pathway Enrichment Analysis of Scans for Positive Selection

In order to summarize the types of loci and explore the potential adaptive genetic architecture implicated by our genome-wide selection scans, we identified all protein coding genes within 100 kb up- and downstream of SNPs showing signatures of selection. Using PANTHER, we then identified pathways that appear enriched in each analysis (Table S7). Significant enrichment for neuro-endocrine signaling and pathways potentially involved in reproduction and thyroid function, including oxytocin receptor mediated signaling, was found for genes near high Pygmy LSBL SNPs (Table S7A). Oxytocin plays a role in lactation, parturition, and pair bonding [30] whereas the TRHR signaling



Figure 3. Genome-wide clusters of signals of natural selection in Pygmies. Blocks of high genome-wide density based on SNPs/Mb for  $F_{ST}$  (top half of each chromosome) and Pygmy LSBL SNPs (bottom half of each chromosome) are shown. Also shown are XP-EHH signals >5.0 in the Pygmy (Black) and Bantu (Red). The only genomic location showing all three signals occurs on chromosome 3 roughly between 45 and 60 Mb. iHS results for this region are given in Figure 4C and 4D. doi:10.1371/journal.pgen.1002641.g003

pathway plays a role in regulation of the hypothalamic-pituitarythyroid axis and influences growth, thermo-regulation, reproduction, and immune response [31]. The observation of a relatively low incidence of goiter in Eastern [32] and Western Pygmies (personal communication, B. Hewlett and L. Cordes) relative to neighboring non-Pygmy populations raises the possibility that Pygmies possess a biological adaptation to a low iodine environment [32,33], consistent with the observation that genes involved in thyroid function are targets of selection in Pygmies [33]. Though there is little significant pathway enrichment for genes near Pygmy-specific XP-EHH and iHS signals (Table S7B-S7C) after correction for multiple tests, several suggestive enrichments are observed for pathways that play a role in immunity and neuroendocrine function. Several of these enrichment categories overlap across different analyses (e.g. 5HT2 type (serotonin) receptor mediated signaling, oxytocin receptor mediated signaling, and TRHR signaling pathways). Finally, a more formal test for enrichment of selection signals at SNPs near genes in the candidate HGH/IGF1 pathway [34] indicated a significant enrichment (p<0.0001) of iHS signals. Similar analyses for  $F_{\rm ST},$ LSBL, and XP-EHH were non-significant.

# Signatures of Selection and Association at GWAS Significant SNPs for Height in Non-Africans

A total of 112 of the 211 SNPs significantly associated with height variation in recent GWAS in non-Africans [12] are also present on the 1M chip, passed QC, and are polymorphic in our study populations (Table S8). Interestingly, the highest  $F_{\rm ST}$  calculated for a non-African GWAS SNP in our study was 0.23, a value far below our

high  $F_{ST}$  cutoff, and most (N = 94) showed  $F_{ST}$  values<0.10. Only 5 show values greater than the approximate worldwide average of 0.15 and only 7 occur within 100 kb up- and downstream of a high FST SNP in our study populations. Similarly, the highest Pygmy-specific LSBL value for these markers was 0.14, also far below our 0.1% cutoff. In addition, the maximum absolute value XP-EHH score for a non-African GWAS significant SNP was 2.68, much lower than our cut-off of 4.08. However, one non-African GWAS significant SNP shows an iHS score in the 0.1% most extreme values in the Pygmies (rs16964211 = 3.233). The positional candidate associated with this SNP is CYP19A1, a member of the cytochrome P450 superfamily. Overall, these findings are consistent with Pickrell et al [35], which reports little evidence for selection at these loci in Western and Eastern Pygmy populations. Additionally, only four of these GWAS significant SNPs show raw p-values for association with stature less than 0.05 in our combined sample (rs1351394 = 0.01978097, rs724016 = 0.02810565, rs9969804 = 0.04384069, rs4630309 = 0.04465145).

Because of the possibility that the European GWAS associated markers and nearby functional variants identified in non-Africans are unlinked in our African populations, we also compared the positional candidate genes reported in non-African GWAS (rather than the SNPs themselves) to a list of all genes within 100 kb windows up- and downstream of signatures of selection in our study. We observe 69 candidate genes for stature that fall within such windows, including *DOCK3* (33 occur near Pygmy-specific signatures of selection, Table S9). However, 50 candidate genes from GWAS were observed in 2000 randomly sampled, genomewide 200 kb bins. Thus, observing 69 such loci in the



**Figure 4. Signals of positive selection and association with height on chromosome 3 between 45 and 60 Mb.** A) Results from  $F_{ST}$  (Black) and Pygmy LSBL (Red) analysis. The region between ~50 and 52 Mb contains the largest concentration in the genome of SNPs with scores in the 0.1% tail of the empirical distribution and contains the positional candidate genes *CISH, MAPKAP3*, and *DOCK3*. B) Results from XP-EHH analysis. Scores depicted in red show negative values indicating a signal in the Bantu samples, and scores in green show positive values indicating a signal in the Pygmy samples. C) Results from iHS analysis in the Bantu samples. D) Results from iHS analysis in the Psgmy samples. E) Results of association analysis for height (cm) using EMMAX with sex as a covariate. Two suggestive associations are evident. The first, at ~46.5 Mb, occurs within the coding region of *LRRC2*. The second is a strong enrichment of association signals between ~49.5 and 51 Mb directly corresponding to Pygmy-specific signals of positive selection in A–D, centered on the three positional candidates *CISH, MAPKAP3* and *DOCK3*.

approximately 3500 such bins identified in our analyses does not appear exceptional. These results suggest that the genetic architecture of height in Pygmies differs from that observed in Europeans; however some overlap in specific genes involved remains a likely possibility.

#### Admixture Mapping in West African Pygmies

In order to incorporate ancestry information directly into association testing, we performed admixture mapping [22] using EMMAX. Specifically, we tested high confidence ancestry blocks inferred in our Pygmy samples (N = 57) by SupportMix (unpublished data) for association with height variation. This analysis identified several suggestive associations with stature (p<0.002; Table S10), though none show striking FDR-corrected p-values. The strongest (p =  $1.40 \times 10^{-5}$ ; FDR = 0.304) is observed for a 50 SNP ancestry bin on chromosome 2 approximately 268 kb from the gene *APOB*. Interestingly, studies have demonstrated a correlation between high *APOB* levels and short stature [36].

# Targeted SNP-Based Association Analysis for Height in Western African Pygmies

In an approach adapted from Moore et al. [37], we next performed targeted association tests, first in Pygmy individuals only, for SNPs showing Pygmy-specific signatures of selection (extreme high Pygmy LSBL, Pygmy XP-EHH and Pygmy iHS; N = 2288 SNPs). These marker-based association analyses identified several suggestive associations with height (Table S11A), though they were not significant after multiple tests correction. The marker showing the strongest association (p = 0.0004;FDR = 0.241) is found in a cluster of immunoglobulin lambda loci. The second strongest association (p = 0.0005; FDR = 0.241) occurs near the gene NDUFA4. While a direct connection to height variation is not obvious for NDUFA4, it is a component of the mitochondrial respiratory chain, which plays a role in metabolic function. The next several strongest associations are found in or near the gene MAP3K2, which is involved in NF-kappa B signaling and the regulation of *JNK* and *ERK5* pathways which play a role in growth factor signaling. Additionally, five of the strongest association signals are observed for markers within the 49–53 Mb region of chromosome 3 (Table S11A).

Since growth hormone and *IGF-1* signaling are strongly implicated in Pygmy-specific phenotypic variation by extensive physiological studies in African Pygmies, we next tested SNPs 100 kb up- and downstream of genes in *HGH*, *IGF-1*, and *INS* signaling pathways (N = 40,558) for association with height variation in the Pygmy population. The markers showing the strongest associations with height near *GH-IGF1-INS* genes are in or near the genes *GCK* (p =  $6.08 \times 10^{-5}$ ; FDR = 0.591), *DUSP4* (p =  $7.82 \times 10^{-5}$ ; FDR = 0.591), and *IGFALS* ( $1.20 \times 10^{-4}$ ; FDR = 0.591), although none of these associations are significant after correction for multiple tests (FDR<0.05).

# Targeted Association with Height Variation in the Combined Pygmy–Bantu Sample

We next performed association tests in the larger combined Pygmy-Bantu dataset for SNPs showing Pygmy-specific signatures of selection. This both maximizes our sample size and takes advantage of the extensive between-population phenotypic variation. We observed 128 SNPs (Table S11B) showing significant associations after correction for multiple tests (FDR < 0.05) and for population substructure. The strongest significant associations are found in or near the genes NDUFA4  $(p = 1.31 \times 10^{-5}; FDR = 0.007), ATP2B4 (p = 2.73 \times 10^{-5};)$ FDR = 0.007), and *DOCK3* ( $p = 2.99 \times 10^{-5}$ ; FDR = 0.007) (Table S11B, Figure 4E). Interestingly, NDUFA4 is also implicated in the Pygmy-only analysis above. The strongest association signal in DOCK3 consists of 8 linked SNPs spread over  $\sim 268$  kb that are found in the middle of the genomic region on chromosome 3 (45 Mb-60 Mb) displaying multiple signals of positive selection (Figure 4A–4D) and association in the Pygmy population (Table S11A). They also occur  $\sim$ 85 kb from a marker (rs13088462) showing significant association with height variation in Europeans [12]. Two additional SNPs flanking this European GWASassociated SNP by  $\sim 4$  kb on either side (rs4443210 and rs7638732) also show significant association with height variation in our analysis (p = 0.00018 and 0.00023; FDR = 0.013 and 0.013, respectively).

We next tested SNPs 100 kb up- and downstream of genes in *HGH*, *IGF-1*, and *INS* signaling pathways for association with height variation in our combined Pygmy-Bantu samples (Table S12). The strongest associations are found in or near *CISH* ( $p = 2.99 \times 10^{-5}$ ; FDR = 0.199), *LEPR* ( $p = 3.36 \times 10^{-5}$ ; FDR = 0.199), *EEF2K* ( $p = 3.44 \times 10^{-5}$ ; FDR = 0.199), *PRL* ( $p = 6.38 \times 10^{-5}$ ; FDR = 0.323), and *IFNG* ( $p = 8.80 \times 10^{-5}$ ; FDR = 0.397), though none were significant after FDR correction for multiple tests.

# Pathway Enrichment Analysis in Extreme Genome-Wide Association Signals

Though our small sample size limits power to detect significant associations using full genome-wide association analyses (see Figure S3), we explored potential pathway enrichment for genes 100 kb up- and downstream of markers in the extreme tail of the empirical distribution of genome-wide p-values (the lowest 0.1%) (Table S7D). Consistent with the physiological literature in African Pygmies, the top two pathways that are significantly enriched in the combined Pygmy-Bantu association analysis are the protein kinase B signaling and the mitogen activated protein kinase/MAP kinase cascades of the Insulin/IGF pathway (Table S7D). While PANTHER enrichment for these pathways does not reach statistical significance after Bonferroni correction for multiple tests, the likelihood that these two pathways are the two most enriched by chance is very low, as evaluated using a non-parametric re-sampling test ( $p = 2 \times 10^{-4}$ ).

## Discussion

Our analyses suggest that a complex set of demographic and selective dynamics have shaped genetic and phenotypic variation in West African Pygmies and neighboring agricultural groups. For example, the very short average tract lengths of inferred ancestry we observe (average Bantu tract length of 3.1 + / - 4.6 Mb) are strikingly different from those seen in simpler admixture scenarios (e.g. African Americans). The shorter average Bantu tract length we observe appears to reflect the long history of admixture between Western Pygmy and neighboring Bantu populations that has taken place, possibly since the Bantu expansion into the rain forest several thousand years ago ([8]). Although models assuming a scenario of a single recent pulse of admixture have been developed for inferring the time of admixture based on the distribution of admixture tract lengths [38], more complex models will need to be developed to make inferences about ancient admixture events (e.g. >100 generations) which may fluctuate in rate over time [39].

While the increased genetic resolution offered by short tract lengths may eventually prove useful in fine mapping the genetic basis of complex traits, it also complicates the application of standard approaches to ancestry inference. The novel, model-free SVM approach to ancestry inference we apply here performs well compared to standard approaches in admixed Western Pygmy populations. However further refinement of ancestry inference methods designed to deal with complex admixture histories as well as multiple and/or unknown ancestral populations are needed. Additionally, the development of unbiased African SNP genotyping arrays (*e.g.* those not based on SNPs and patterns of LD identified in non-African populations) are likely to provide novel insights into genome-wide patterns of ancestry and, coupled with increased samples sizes, will facilitate association studies of complex trait variation in African samples.

# Signatures of Selection

Our multiple tests of neutrality, which are sensitive to different selection scenarios and timescales, revealed many signatures of positive selection which are enriched for genes influencing a variety of complex traits with potential fitness implications including body size, thyroid function, immune system function, and reproduction. Although we see evidence for enrichment in HGH pathway genes in our Pygmy-specific iHS results, suggesting some selection on these loci after the split between Eastern and Western pygmies (>10 kya), the genome-wide distribution of signals is not consistent with a strong, recent selective sweep at a single locus. Rather, the overall results of our selection scans suggest that the process of local adaptation involved multiple loci that may have been favored at different points in time. Additionally, targets of selection from standing variation or those showing slight changes in allele frequency at multiple loci are challenging to identify. Further development of methods sensitive to these types of complex adaptive signatures will greatly facilitate the analysis of the genetic architecture of adaptation in these and other global populations [40].

## The Chromosome 3:45–60 Mb Region

The region found on chromosome 3 between 45 and 60 Mb shows an overlapping pattern of selection and association signals

(Figure 4). The presence of multiple, Pygmy-specific signals of selection using different tests of neutrality could indicate that the region has been repeatedly under selection during the process of local adaptation in Western Pygmy populations. However, the additional presence of Bantu-specific signatures of selection suggests that the region may be more broadly relevant for adaptation in human populations. Indeed, this region has been previously implicated in genome-wide scans for selection in Hapmap populations from Europe, Asia, and Africa [23,28,29,41,42,43]. SNPs in this region also show significant association with height variation in both our combined Pygmy-Bantu samples and in recent GWAS studies in Europeans as well as suggestive association in Pygmy samples only (Tables S11 and S12).

Among the most promising positional candidates in the chromosome 3 region are DOCK3, a guanine nucleotide exchange factor shown to be associated with height variation in non-African populations [44], MAPKAP3 a mitogen-activated protein kinase central to several signaling pathways [45], and CISH, a negative regulator of cytokine signaling that also inhibits HGHR activity [46]. The 709 kb long DOCK3 gene is expressed specifically in the brain [47] and has no obvious impact on stature, although a SNP within DOCK3 is associated with height in Europeans [12]. However,  $\sim 63$  kb away from *DOCK3*, oriented in the opposite direction, is the 5.4 kb CISH gene. CISH, a member of the cytokine signaling (SOCS) family of proteins, is up-regulated by interleukin-2 (Il-2) and plays a critical role in the signaling of cytokines by binding tyrosine residues on activated cytokine receptors, particularly IL-2R [48], as well as T-cell differentiation [49]. A recent study demonstrated that genetic variation at CISH is associated with susceptibility to bacteremia, malaria, and tuberculosis in several global groups including a Gambian population [48], indicating the important role that CISH plays in infectious disease susceptibility. CISH also directly inhibits HGHR action by blocking the STAT5 phosphorylation pathway [49], and CISH expression is highly regulated by levels of HGH expression [50]. Transgenic mice that over-express CISH show reduced growth and overall small body size [49].

Sequencing of 5 kb of the *CISH* gene inclusive of the protein coding region, (Figure S4A–S4C) reveals increased diversity in Pygmy individuals relative to the Bantu individuals, but no deviations from neutral expectations. We also did not identify any common population-specific SNPs. However, it is possible that a genetic variant upstream of *CISH*, possibly within or near the *DOCK3* gene, contains a regulatory element that alters *CISH* gene expression in Pygmies.

Although we did not detect any structural variants (SVs) in the region using SNP data (data not shown), DOCK3 is known to contain inversion polymorphisms in global populations between  $\sim$  50.88–50.97 Mb [51]. This observation raises the possibility that an undetected SV could play a role in altering gene expression in the region. Indeed, 8 SNPs among the most strongly associated with stature in our combined Pygmy-Bantu analysis occur in the DOCK3 gene and are in 100% LD over a  $\sim$ 268 kb region in both Pygmy and Bantu populations. Specifically, after phasing, genotypes at these SNPs are found as two "scaffold" haplotypes: AAGGGAAG and GGAAACGA, with the "A" form (those with an A at the first SNP, rs6779819) at  $\sim$ 73% frequency in the Pygmy sample, and  $\sim 36\%$  in the Bantu. These markers also show strong linkage disequilibrium in HapMap European and Asian samples where the same "A" form haplotype is found at 26% and 38% frequency, respectively. Additionally in our Pygmy-Bantu analyses, variation at intervening marker loci define lowerfrequency haplotypes, several of which are private (or near private) in the Pygmy population across the region (right half of Figure S5B). This pattern of variation is consistent with the possibility of an ancient inversion encompassing the 8 "scaffold" SNPs that has restricted recombination in the region, with the subsequent accumulation of variation resulting in each sub-type. Moreover, we observe multiple long-range population-specific haplotypes (~1 Mb in size; Figure S5) in the CISH/MAPKAP3/ DOCK3 region in both the Bantu and Pygmy populations that likely contribute to the high  $F_{ST}$  signal (Figure 4A). Additionally, we observed reduced levels of switching between ancestry blocks in the 46 Mb-53 Mb region on chromosome 3 (Figure 2B), consistent with reduced recombination. Future functional studies will be required to determine the magnitude of any CISH expression differences between Pygmies and other populations and how these might alter the downstream effects of HGH in African Pygmies.

In addition to CISH, MAPKAP3, and DOCK3 there are several other positional candidate genes in the chromosome 3 region between 45 and 60 Mb that may have played a role in Western Pygmy adaptation. These include ERC2, at position ~55 Mb, which occurs within 100 kb of signals in all five scans for selection (Table S9). Together with BSN located at ~49 Mb in the chromosome 3 region, ERC2 appears to help regulate neurotransmitter release [52]. This raises the interesting possibility that an interaction between these loci, possibly related to reproduction, is contributing to the strong Pygmy-specific differentiation in the region. Thus, it is possible that a co-adapted complex of genes in the chromosome 3: 45–60 Mb region has been under selection in the Western Pygmy lineage, as has been observed in other species in regions with reduced recombination due to structural variation [53].

## Pathway Enrichment

The pathway enrichment results from our selection scans and height association results provide suggestive, novel insights into the possible genetic and physiological architecture of adaptation in Western African Pygmies. Signatures of selection in Western Pygmies are often found near genes involved in neuro-endocrine signaling pathways related to reproduction and steroid hormone synthesis, supporting the proposal that differences in the timing of reproduction may partly explain short stature as adults in these populations [1]. Further study of specific genes in these pathways and their expression patterns are likely to yield novel insights about their functional role in adaptation in Pygmies. In addition, the enrichment for genes that play a role in the Insulin/IGF pathway near the strongest signals of genome-wide association further supports a major role for this pathway in the genetic architecture of height in Western African Pygmies.

#### Conclusions and Future Studies

Our multi-dimensional approach incorporating ancestry estimation, genome-wide scans for positive selection, and genotype/ phenotype association identified several candidate genes and pathways that may contribute to adaptive phenotypes, including short stature, in Western African Pygmy populations. Our results raise the possibility that the adaptive process that produced small body size in Pygmies may be the result of selection for traits other than stature, including early reproduction, metabolism, and immunity, and that the functional variants (possibly regulatory in nature) that are targets of selection may have pleiotropic effects. Indeed, given the extremely high pathogen load and short life span in Pygmies (between 15.6 and 24.3 years [1]) immune function is likely to have been under strong selective pressure in Pygmy populations. Furthermore, many of the loci that play a role in immune response (e.g. cytokines) are known to directly influence genes related to HGH and insulin/insulin-like growth factor signaling [54]. It is also possible that there have been multiple selective events in the history of the Pygmy populations, at different time periods, that may have contributed to the coadaptive evolution of loci that play a role in immunity, metabolism, and neuro-endocrine function. Comparison of selection and association signals in Western and Eastern Pygmies will be informative for distinguishing if selection for short stature evolved in an ancestral Pygmy population prior to their divergence within the past  $\sim 25,000$  years [3,4,5,6], or whether this trait evolved independently in these populations due to convergent evolution. In addition, future analysis of whole genome sequence, characterization of structural variants, and functional studies of gene expression after challenge with HGH and IGF-1 in appropriate cell types will be particularly informative for identifying genomic variants that play a role in short stature in Pygmy populations.

## **Materials and Methods**

#### Subjects and Samples

Appropriate IRB approval for this project was obtained from both the University of Maryland and the University of Pennsylvania. Prior to sample collection, informed consent was obtained from all research participants, and permits were received from the Ministry of Health and National Committee of Ethics in Cameroon. In total, our study included 132 individuals from six populations residing in Cameroon. Three are Niger-Kordofanian Bantu-speaking hunting and gathering populations: the Baka, Bakola, and Bedzan, and three are neighboring Niger-Kordofanian Bantu-speaking agricultural populations: the Ngumba, Southern Tikar, and Lemande. While the term 'Pygmy' has historically been pejorative, more recently it has been used by indigenous groups themselves as well as activist groups working on their behalf [55,56,57]. In acknowledgement of this recent trend and in the absence of a better term that encompasses the hunting and gathering peoples from Cameroon, we use 'Pygmy' to collectively refer to the set of three populations, the Baka, the Bakola, and the Bedzan, included in our study and their ancestors. Likewise, we will use the term 'Bantu' to refer to the three agricultural populations in our study, the Lemande, the Ngumba, and the South Tikar, and their ancestors. Samples were collected in villages in the Eastern Province (Baka and neighboring Bantu groups), Southern and Ocean Provinces (Bakola and neighboring Bantu groups), and Center Provinces (Bedzan and neighboring Bantu population). White cells were isolated in the field from whole blood with a salting out procedure modified from [58] and DNA was extracted in the lab with a Purgene DNA extraction kit (Gentra Systems Inc., Minneapolis, MN). Height measurements as well as self-identified ethnicity, parent and grandparent information were also recorded except in the case of the Lemande for which no phenotypic measurements are available.

# Genotyping and Quality Control

DNA samples were genotyped on the Illumina Human 1M-Duo DNA analysis BeadChip. All DNA samples showed SNP call rates of at least 95%. In order to ensure the quality of genotype calls, SNPs with more than 5% missing data were excluded from analysis. The remaining autosomal dataset includes a total of 1,083,730 SNPs of which 960,497 were polymorphic in our populations of interest. An additional 40,949 X chromosome SNPs exceeded similar quality control metrics and were analyzed separately to determine  $F_{\rm ST}$  and Pygmy LSBL values. In order to identify close genetic relatives in the autosomal dataset, we estimated identity by descent (IBD) using PLINK v 1.07 [59]. Seven individuals (4 Bakola, 2 Bedzan, and 1 Tikar South) were closely related to one or more individuals in the sample. After their removal no pairwise estimate of pi-hat exceeded 0.25, a value roughly corresponding to relationships closer than grandparent-grandchild. The remaining set of 125 samples (67 Pygmies and 58 Bantu) served as the basis for subsequent analyses.

## Estimates of Population Structure and Ancestry

In order to evaluate population structure in our sample, we identified a subset of SNPs that maximizes the number of markers while minimizing linkage disequilibrium (LD). Specifically, we removed SNPs within a sliding window (100 SNPs per window and an offset of 50 SNPs) based on the variance inflation factor [59] (VIF = 2). The resulting 30,404 markers were analyzed using STRUCTURE to estimate the proportion of individual ancestry for a range of ancestral clusters, K = 2, 3, and 4 (Figure 1A). Principal components analysis (PCA) was performed using R (Figure 1B, 1C; R Development Core Team [60]).

# Correlation between Global Ancestry and Height Variation

We explored the relationship between global admixture and height variation using linear regression. Estimates of ancestry were analyzed using the following model:

$$Y_{ij} = \mu + \beta_1 G_j + \beta_2 P_j$$

Where G is a co-variate representing sex and P represents the proportion of Pygmy ancestry. Age was initially included but was not significant and so was removed from the final analysis (Figure 1D).

#### Pairwise F<sub>ST</sub> and LSBL analyses

First, using the subset of least-admixed Pygmy (N = 28) and Bantu (N = 40; Baka = 17, Bakola = 13) individuals identified in our STRUCTURE analysis, we calculated pairwise  $F_{ST}$  [24] values for all SNPs that passed quality control. Cross-population scans for selection (e.g. FST and LSBL) are expected to be useful for detecting regionally restricted adaptation, including selection on standing variation [26]. SNPs with FST values greater than 0.3308 represent the 99.9% tail of the empirical distribution and are referred to as "high FST SNPs". Next, in order to identify allele frequency differences specific to the Pygmy lineage, we incorporated data from unrelated individuals from the HapMap Phase 3 Maasai sample (N = 45) [61]. We then calculated Pygmy locusspecific branch length (LSBL) values [25] for each SNP that appeared in both datasets. Values greater than 0.2215 represent the 99.9% tail of the empirical distribution and are referred to as "high Pygmy LSBL SNPs". In order to better visualize both the clustering and concurrence of F<sub>ST</sub>/LSBL signals, we calculated and plotted the density of high FST and high Pygmy LSBL SNPs per Mb for all non-overlapping 500 SNP windows across the genome (Figure 3). A matrix of average pairwise FST values between each population is given in Table S13. We were also interested to see if any unusual patterns of population structure exist on the X chromosome, and so analyzed these data separately. The Illumina software package genome studio (Illumina, Inc, San Diego, CA) was used to call SNPs in the female individuals only, and once this was accomplished, males were re-incorporated to establish hemizygous allele calls.

## **XP-EHH and iHS Analyses**

Using binary executables made publically available by the Pritchard laboratory at the University of Chicago, we performed both cross-population extended haplotype homozygosity (XP-EHH) [29] and integrated haplotype score (iHS) [28] tests of neutrality. The XP-EHH test is most powerful when the selected allele and its associated haplotype are nearly fixed in one population but remain polymorphic or absent in a comparison population. In contrast, iHS is most powerful when a selected haplotype has reached intermediate frequency and is thus useful for detecting partial selective sweeps within a single population [29]. The iHS test has also been shown to be more robust than alternative methods under complex demographic histories and variable recombination rates [62].

SNP genotypes were computationally phased using fastPHASE v 1.4 [63] treating each chromosome independently. Haplotypes that minimized the fastPHASE switch error were used in subsequent analyses, as is recommended for data sets with many markers [63]. A fine-scale recombination map relevant to the African populations included in our study was generated using LDhat version 2.1 [64] and a dataset including a total of 100 unrelated samples. Specifically, we analyzed 25 males and 25 females, from each of two populations in HapMap3 Release 2 [65]: the Yoruba from Ibadan, Nigeria (YRI) and the Luhya from Webuye, Kenya (LWK). HapMap3 Release 2 (January 2009) data were used rather than the more recent HapMap3 Release 3 (May 2010), since these were phased using trios.

LDhat uses a Monte Carlo approach to sample from the posterior distribution of the population genetic parameter  $4N_cr$  for intervals between consecutive markers, where  $N_e$  is the effective population size and r is the per-generation recombination rate. The LDhat Markov chain was run for 10,000,000 iterations; 1,000,000 iterations were used as burn-in and only every 5,000<sup>th</sup> sample was retained to reduce auto-correlation among the posterior samples. This was done for both populations separately. The median of the retained samples was calculated in order to obtain point estimates for the population-specific  $4N_e r$  values. To avoid any population-specific demographic or selection effects in the recombination map, the YRI and LWK estimates were averaged. Using these population-averaged estimates, we computed genetic positions for each SNP in our data in units of  $4N_cr$  that were subsequently used in the XP-EHH and iHS tests. For SNPs present in our data but not in HapMap3, genetic distance was assumed to scale linearly within intervals.

iHS analysis requires ancestral state information. To establish ancestral and derived alleles for all SNPs in our data, we used the human, chimpanzee, orangutan, and rhesus macaque reference alleles in dbSNP131 [66]. The data were downloaded from table snp131OrthoPt2Pa2Rm2 of the University of California at Santa Cruz genome browser [67]. An allele was determined to be ancestral to the human population if the human reference matched 1) the corresponding chimpanzee allele, or 2) the orangutan allele if the chimpanzee allele was missing, or 3) the rhesus macaque allele if both the chimpanzee and orangutan data were missing. Approximately 5% of the SNPs in our data could not be assigned an ancestral state, due mainly to missing data in dbSNP131; this was consistent across all autosomes (data not shown). SNPs with minor allele frequencies less than 5% in either the Pygmy or Bantu populations were removed from the phased dataset in agreement with recent publications [e.g. 28]. SNPs with ambiguous ancestral states were also removed from the iHS analysis. A single XP-EHH process comparing the Pygmy sample to the Bantu samples was executed and two independent iHS processes, one for the Pygmy and one for the Bantu samples, were

performed. The non-standardized scores returned by the XP-EHH and iHS binary executables were adjusted such that all scores had zero means and unit variances, either with respect to the entire data set (for XP-EHH, as described in reference [29]) or to SNPs with similar derived allele frequencies (for iHS, as described in [28]). The threshold values established from the 99.9% tail of each empirical distribution were as follows: absolute value of XP-EHH = 4.08, iHS\_Pygmy = 3.18, and iHS\_-Bantu = 3.16. This imposes cutoffs that are either comparable or more conservative compared to the values used in previous studies using these tests (e.g. [29]; [28]). It is important to note that genomic regions identified using "outlier" approaches are not all expected to be targets of selection. Rather, SNPs within the tails of the empirical distribution are expected to be enriched for linkage with adaptive variants or can be false positives caused by historical demographic events [29]; [28]. We focus primarily on Pygmyspecific signals as these are of greatest interest with respect to identifying candidate loci that play a role in short stature in that population.

#### Local Ancestry Inference and Admixture Mapping

Common methods for admixture mapping assume recent admixture, relatively long tracts of LD throughout the genome, and the presence of both reference parental population samples [68]. While methods such as STRUCTURE are able to estimate admixture proportions well globally, they perform poorly with respect to the local ancestry assignments critical to accurate admixture mapping analysis [69]. Other methods such as LAMP [69] do significantly better in this respect, but require knowledge of allele frequencies from both ancestral populations for optimal performance [70]. HAPMIX [71] and PCAdmix [72] have similar requirements. In addition, with the exception of HAPMIX, most methods require unlinked markers. The demographic history of Western African Pygmy populations creates a unique challenge for these methods since admixture is both ancient and on-going. Thus, we expect a wide distribution of haplotype sizes including many that are relatively short. Furthermore, since every Pygmy individual in our sample shows some degree of Bantu admixture, only one of the two ancestral populations is still available and sampled (Bantu). We therefore have applied a novel method of local ancestry inference, SupportMix, that can identify relatively small tract lengths to estimate genome-wide ancestry (unpublished data). Simulations demonstrate this method is more accurate than LAMP for several different degrees of population structure (unpublished data). When LAMP-ANC was performed using information from both ancestral populations and compared to SupportMix using only a single ancestral source population, SupportMix was again more accurate (Figure S6). Thus, SupportMix is an appropriate method to use when ancestral allele frequencies are known for only one ancestral population, as is the case for the Western Pygmies.

Briefly, SupportMix is a two-step process that classifies the ancestral origin of small regions of the genome using a support vector machine (SVM) trained on the ancestral population(s). This is followed by a classification of ancestry with the aid of a Markov model. Specifically, the SVM learns the parameters of the Markov-process and the most probable ancestral state is solved by the forward-backward algorithm over the observed states. As noted above, the complex pattern of admixture observed in our Pygmy samples hinders this training step. To overcome this, an additional SVM was trained at each window on all Pygmy and Bantu samples. Those Pygmy samples most different from the Bantu as measured by the L2 norm from the dividing SVM hyperplane, were chosen as region-specific "ancestral" Pygmies. This

allowed a new support vector machine to be trained to segregate these region-specific "ancestral" individuals from the Bantu. This was then used to classify the remaining Pygmy samples as either ancestrally Bantu or Pygmy for a given genomic region. The proportion of region-specific "ancestral" Pygmies, chosen in each window was the same for every window and was chosen to obtain a global ancestry proportion matching that of STRUCTURE. This agreement is seen in Figure S7A. A window size of 50 consecutive SNPs was sufficient to separate the "ancestral" Pygmies from the Bantu leading to a high posterior probability (69% of windows above 0.9) from the Markov Process. Only these "high confidence" regions of inferred local ancestry were used in subsequent admixture mapping analyses.

### Association Analyses

Association analyses using this dataset are complicated by the inclusion of individuals from two differentiated populations, the complex patterns of admixture within the Pygmies, and our small sample size. These factors greatly reduce the power of a genomewide association approach (Figure S3). Instead, we used a combination of ancestry-based and marker-based approaches that are expected to produce non-overlapping but complementary information [73] within and between populations. We correct for multiple tests using an analysis-specific false discovery rate (FDR) approach [74,75] and interpret the most extreme association signals as suggestive. Specifically, FDR values were calculated from the vector of p-values from each association analysis separately. First, we performed admixture mapping using high confidence ancestry blocks inferred by SupportMix in our Pygmy samples (N = 57). Second, in an approach adapted from Moore et al. [37] we performed association tests for two subsets of SNPs for which there are a priori expectations of association in Pygmy samples separately. These are 1) SNPs showing Pygmy-specific signatures of selection (extreme high Pygmy LSBL, Pygmy XP-EHH and Pygmy iHS; N = 1823 SNPs) which are expected to be enriched for SNPs in linkage disequilibrium with functional variants that play a role in recent adaptation in Pygmies and 2) SNPs 100 kb up- and downstream of genes in HGH, IGF-1, and *LNS* signaling pathways (N = 40,558) which are strongly implicated in Pygmy-specific phenotypic variation by the extensive physiological studies in African Pygmies described above. Genes with roles in GH, IGF-1 and INS signaling were defined via STRING using the criteria: highest confidence, no more than 50 interactors, 200 additional nodes. Third, we performed similar marker-based analyses using all Bantu and Pygmy samples combined in order to maximize both phenotypic and genetic differentiation as well as sample size (N = 125). Finally exploratory genome-wide association analyses were performed using Pygmy samples only, Bantu samples only and the combined dataset (Figure S3).

In all cases, genotype/phenotype association analysis was conducted using EMMAX [76], a mixed-model linear regression approach that corrects for population structure even under cases of extreme differentiation [77,78]. This method has been successfully used for association mapping in a sample of inbred, and highly sub-structured, dog breeds [77]. Simulations also demonstrate that EMMAX performs extremely well in correcting for population substructure [78]. The additional advantages of this mixed-model approach include simultaneous correction for both relatedness within populations and structure between them via a pair-wise matrix of genetic relationships among individuals. We obtained results using both matrices included in the current distribution of EMMAX (BN and IBS) for both SNP-based and admixture-based analyses. Since they were very similar, we focus on results obtained using the IBS matrix for simplicity. In all analyses, but especially the combined Bantu and Pygmy analysis, EMMAX appears to have provided a reasonable correction for population structure as judged by the resultant Quantile-Quantile plots and the generally low correspondence between high LOD scores and high  $F_{ST}$  (Figure S8). Further, by overlaying both *a priori* biological information as well as results from several tests of neutrality our approach, in effect, examines the joint distribution of several test statistics, a scenario under which power is expected to be greatly enhanced.

## Pathway Enrichment Analyses

Several gene lists were generated based on the analyses above and PANTHER was used to summarize and explore types of loci and pathways that are enriched in the tails of the empirical distribution for signals of selection and association. Specifically, genes residing 100 kb up- and downstream of each high F<sub>ST</sub> and high Pygmy LSBL SNP, as well as those with XP-EHH and iHS scores in the 99.9% of the empirical distribution were identified (100 kb flanking windows were chosen since targets of selection could include regulatory regions). Each was submitted to the PANTHER web tool separately (Table S7A-S7D). We also performed similar enrichment analysis for the lowest 0.1% of the empirical distribution of genome-wide p-values from marker-based association analyses (p < 0.001; Table S7A) as well as all pvalues<0.05 using ancestry estimates from SupportMix (Table S7B). Pathways passing the significance threshold for analysisspecific Bonferroni corrections for multiple tests are given in bold. In addition, we calculated the probability that the two Insulin/IGF signaling pathways would be the two most enriched in the genes 100 kb up- and downstream of markers in the extreme tail of the empirical distribution of genome-wide p-values (the lowest 0.1%) for association (Table S7D). Specifically, we sampled random sets of genes with HUGO symbols (N = 1248, corresponding to number of genes in the empirical data); only genes with UNIPROT identifiers were retained. UNIPROT identifiers were mapped to all pathways using the Panther database (ftp://ftp. pantherdb.org). We then calculated chi-squared statistics for each pathway in each random list and recorded which showed the two most extreme test statistics. We repeated this procedure 10,000 times and identified either the Insulin/IGF pathway-protein kinase B signaling cascade or the Insulin/IGF pathway-mitogen activated protein kinase kinase/MAP kinase cascade as one of the two most enriched only twice  $(p = 2 \times 10^{-4})$ 

In order to more directly quantify the potential enrichment of selection and association signals in HGH pathway genes, we used an approach described by Eleftherohorinou et al [34]. Specifically we generated a cumulative test statistic,  $CA_{test} \sum_{s \in HGH} \beta_g$ , where s is the index of all SNPs in the *HGH* pathway and  $\beta$  is the teststatistic (score) estimated for each SNP in a given analysis. If there were an enrichment of signals in the HGH pathway, we would expect  $CA_{test}$  to significantly deviate from 0. To assess this, we used a genome-wide re-sampling approach wherein the same number of SNPs found in the HGH pathway was randomly sampled with replacement from our set of genome-wide SNPs. We then calculated a new cumulative statistic CArand. This procedure was repeated 10,000 times to create a null distribution for each analysis. A skew-normal distribution was fit to the null using maximum likelihood, as described in Eleftherohorinou et al. [34]. A p-value for the observed statistic was calculated using the best-fit null skew-normal distribution.

#### CISH Re-Sequencing

We re-sequenced a 5.4 kb region encompassing the CISH gene and its promoter in 19 Bantu (6 Lemande, 7 Tikar South, and 6 Ngumba) and 23 Pygmy (9 Baka, 8 Bakola, and 6 Bedzan). We first amplified the entire region using the two primer sets: 5'-CCTAGAGGGTCACCTATAACCTACAC-3', forward 1 reverse\_1 5'-CTTGCTGCTTATCCTCGTCCTTAC-3' and forward\_2 5'-CCTCTGAGAGACACTCCTATCCAT-3', reverse\_2 5'-CCTGCTGTCTATCCTCGTCCTTAC-3' (Figure S4A, S4B). Individual PCR amplifications were visualized with electrophoresis in a 1% agarose gel and sequenced with the Big Dye Terminator cycle sequence kit (Applied Biosystems, Foster City, CA) and run on an ABI prism 3730XL (Applied Biosystem, Foster City, CA,). For primers used for sequencing see the legend of Figure S4. We performed additional sequencing of a 1.2 kb region to evaluate the reported functional single nucleotide polymorphisms -639, -292 and -163, described by [48] using the primers: forward 5'-GTCCGCATAACGGGAGCAACAC -3' and reverse 5'-CGCTTACCCCTGAACGCAGAGGACC -3'. CISH sequence results were analyzed and evaluated with the software package Sequencher (http://www.genecodes.com/). Haplotypes were estimated with the software package Phase [79,80,81]. Network analysis was performed using the program Network (http://www.fluxus-engineering.com) using the median joining methodology (Figure S4C) [82]. DNAsp (http://www.ub. edu/dnasp/) was used for calculating nucleotide diversity and Tajima's D (Figure S4D) [83].

# **Supporting Information**

**Figure S1** Phenotypic variation for height. Height distributions in A) the combined Pygmy-Bantu sample, B,C) males and females separately and D,E) ethnic groups separately. (EPS)

**Figure S2** Global frequencies of rs7626978. World-wide frequencies of rs7626978, the most structured SNP in the Chromosome 3: 45–60 Mb region that was also genotyped in the Human Genome Diversity Panel (HGDP). Despite the fact that this SNP was likely identified in the Hapmap Yoruba population, it is most common in Pygmy and San hunter-gatherer and neighboring populations. (EPS)

Figure S3 QQ plots for within-Population Genome-wide Association Using EMMAX. Genome-wide QQ-plots of association results for A) Pygmy samples only and B) Bantu samples only. A genome-wide approach is expected to be underpowered given our limited number of samples. Indeed, no genome-wide significant associations are observed in Pygmy samples only. However, analysis of unrelated Bantu individuals only (N = 39)with height data) yields both a genome-wide significant association  $(1.31 \times 10^{-8})$  and a seven-fold enrichment of p-values less than  $10^{-6}$  (expected = 941,183 polymorphic markers  $\times 10^{-6} = 0.941$ ; observed = 7, reduced from 10 due to linkage). The genome-wide significant association appears at three tightly linked SNPs (rs13627, rs2189526, rs3815170) located on the short arm of chromosome 19 within the coding region of the gene GRIN3B, an ionotropic glutamate receptor. While this region is not identified in the FST, LSBL or XP-EHH results, it does show evidence of an extended haplotype in Pygmies using iHS. The strong statistical signal of association results from the 8 shortest Bantu individuals being heterozygous for the three markers. Interestingly, marker genotypes are at roughly equal frequencies in both Pygmy and Bantu samples. One possible interpretation of this result is that a short-stature haplotype from the Pygmy population is segregating in our Bantu sample. However, the eight individuals show varying haplotypes in the 100 SNP windows surrounding GRIN3B. Future re-sequencing of this candidate gene, however, may well reveal a protein coding or regulatory variant with functional consequences. We also observe a suggestive association near *GRIN3B* in the marker-based Pygmy/Bantu association analysis ( $p = 1.10 \times 10^{-5}$ ) and we observe a suggestive association ( $p = 1.46 \times 10^{-4}$ ) near GRIN3A in the ancestry-based analysis. These results raise the possibility that glutamate signaling could play a role in stature in these populations. (EPS)

Figure S4 Re-sequencing results for CISH. Sequencing of the PCR amplification product was performed with the following primers: F\_seq\_I: GGTTGGTGGGACCCTTTATT; F\_seq\_II: CTCCACTCACTGCTCATCCA; F\_seq\_III: GGCAGAAGG-CACGTTCTTAG; F\_seq\_IV: CTGACTGTACGGGGGCAA-TCT; F\_seq\_V: ATTGTTGAGGGGTAAG; F\_seq\_VI: CCA-CTGCAGTTCTGCTAGGTC; F\_seq\_VII: CCAAATCGCA-AGTGGAGAAT; F\_seq\_VIII: ACCGGACAGTCTGACTTT-GG; R\_seq\_I: ACCAGCCCTGACTTCTCTGA; R\_seq\_II: C-CCCATTGAGGGCTCTGTA; R\_seq\_III: GCCTTGGTTCC-AGCAAGATG; R\_seq\_IV: CTTGCTCTTGCTGGCTCTTC; R\_seq\_V: TGACAGCGTGAACAGGTAGC; R\_seq\_VI: GC-CACCTTGATTGTTTCCAT; R\_seq\_VII: GGTGGACAGT-GGCCTCTAAA; R\_seq\_VIII: GGCTGGGGAAGATGACAC-TA; F\_seq\_v: TGCAGGACTCTCTCTCAGCA; F\_seq\_w: GA-GTTCCACCGCGAGATAAG; F\_seq\_x: CACTGCCTCTC-AGTCCCTGT; R\_seq\_v: CTCTTATCTCGCGGTGGAAC; R\_seq\_w: TGAACGCAGAGGACCATGT. A) Gene structure of the re-sequenced region. B) SNP genotype calls for each sequenced individual. C) Phased haplotype network for the resequenced region. Circles depict haplotypes and their relative sizes represent observed frequencies. Yellow indicates the proportion of each found in the Bantu and blue the proportion found in Pygmies. The black circle in the lower right represents the chimpanzee sequence. Red numbers along edges correspond to segregating sites that differ between nodes. Mv1-mv6 are variants not present in the sample. No significant differences in height were detected for any set of haplotypes. D) Nucleotide diversity and Taijma's D for the combined dataset as well as by population. (EPS)

Figure S5 Visualization of fastPHASE-inferred haplotypes. Plots exclude SNPs with minor allele frequencies less than 0.1 and are centered on rs107457, the marker showing the highest LSBL value in the chromosome 3 region of interest. A) unsorted visualization B) sorted by haplotype within Pygmy and Bantu groupings. In order to visualize the haplotype structure surrounding SNPs of interest, a custom script was written in R to plot exact haplotype matching for fastPHASE-inferred haplotypes centered on a given SNP. Markers with minor allele frequencies less than 0.1 were excluded. Given a specified SNP of interest and window size (in number of SNPs up- and downstream), the top 10 most frequent haplotypes in the window were identified. These "templates" were assigned the following colors from most to least frequent: black, grey, red, orange, yellow, dark green, light green, dark blue, blue, and turquoise. Next, building outward from the specified SNP, each haplotype in each individual was matched to one of the 10 templates and colored accordingly. Haplotype matches are assessed one SNP at a time first to the left and then to the right of the marker of interest. In the event that an individual haplotype matches multiple templates over the interval at hand, the color representing the most frequent in the window as a whole was retained. Any haplotype stopped being assigned a color at the point that it failed to match one of the templates.



Figure S6 Comparison between SupportMix and LAMPANC using simulated admixture. In-silico admixed HapMap3 individuals between Yoruba (YRI), French (CEU) and Yoruba, Maasai (MKK) were examined with SupportMix using only the Yoruba as an ancestral population. Both ancestral populations were used in LAMPANC. The error bars represent one standard deviation in accuracy across the 25 individuals tested. Specifically, 25 individuals of Yoruba and French ancestry and 25 individuals of Yoruba and Maasai ancestry were generated in-silico using the phased HapMap3 data. The admixture was generated using the same method as described in<sup>32</sup>. SupportMix was carried out exactly as is described the main text and LAMP was run in LAMPANC mode using the allele frequencies of both ancestral populations. It is clear that even when LAMPANC is given more information (allele frequencies for both ancestral populations) it is less accurate in calling ancestry than SupportMix. (EPS)

Figure S7 Results from SupportMix analysis of local ancestry. A) Average Pygmy ancestry per individual as estimated by SupportMix and STRUCTURE. The STRUCTURE estimates were obtained from a thinned version of the SNP dataset. B) Plot of ancestry tract lengths as obtained by SupportMix with Pygmy tracts averaging 1.7+/-2.4 Mb and Bantu tracks 3.1+/-4.6 Mb. The slight hump at 120 kb represents the average window length and represents a lower bound on the possible tract lengths observable with SupportMix. C) Quantile-Quantile plot of pvalues from admixture mapping for Pygmy height obtained from ancestry assignments using SupportMix and EMMAX corrected for population structure using an identity by state (IBS) kernel matrix. D) Genome-wide p-values from admixture mapping for Pygmy height. Each dot represents the p-value as obtained from associating height on ancestry for a window of 50 consecutive SNPs using EMMAX corrected with a kernel matrix representing IBS

## (EPS)

Figure S8 Performance of EMMAX in correcting for population structure. A) QQ plot depicting the severe, upward bias in genome-wide  $-\log 10(p)$  values when no correction for population structure is applied. B) QQ plot of -log10(p) values from a fixedeffect linear regression including overall STRUCTURE estimate of Pygmy ancestry to correct for population structure. C) OO plot of genome-wide results obtained via EMMAX showing excellent control of p-value inflation due to population structure. D) Hexagonal binning plot of the relationship between p-value and  $F_{ST}$  in the genome-wide results from EMMAX. SNPs showing the lowest p-values (surrounded by the black box) are essentially evenly distributed over the full range of  $F_{ST}$  values. There also appears to be a qualitatively higher density of low p-values for lower values of F<sub>ST</sub> further indicating a biologically negligible relationship between FST and association with height. E) Table showing the low correlation of F<sub>ST</sub> with p-value both genome-wide and in the subset of high F<sub>ST</sub>/LSBL SNPs. F<sub>ST</sub> and p-values for SNPs tested for association using EMMAX are only very weakly correlated (Spearman's Rank correlation genome-wide = -0.14; Pearson's correlation coefficient of -0.17, -0.2 (r) and -0.22(rho) for high F<sub>ST</sub> and high LSBL SNPs only). Thus, we do not see a strong correlation between SNPs with high differentiation and low p-values in association tests, as expected if population structure alone were driving the global enrichment for low p-values. (EPS)

 with height using the IBS matrix from EMMAX in the combined Pygmy-Bantu sample and analysis-specific Benjamini-Hochberg FDR corrected p-values are given. (XLSX)

**Table S2** High Pygmy LSBL SNPs. Position information, LSBL and association results for the top 100 SNPs in the empirical distribution of autosomal LSBL values. Raw p-values for association with height using the IBS matrix from EMMAX in the combined Pygmy-Bantu sample and analysis-specific Benjamini-Hochberg FDR corrected p-values are given.

(XLSX)

 $(\mathbf{XLSX})$ 

**Table S4** Extreme XP-EHH SNPs. Position information, IHH scores for both Pygmy and Bantu groupings, raw and standardized XP-EHH scores and association results for the top 100 SNPs in the empirical distribution of autosomal XP-EHH values. Raw p-values for association with height using the IBS matrix from EMMAX in the combined Pygmy-Bantu sample and analysis-specific Benjamini-Hochberg FDR corrected p-values are given. (XLSX)

**Table S5** Signals of selection from iHS analysis in Pygmy samples. Position information, iHS score, the number of SNPs on which it is based, and association results for the top 100 markers showing scores in the empirical distribution of autosomal iHS scores in Pygmy individuals. Raw p-values for association with height using the IBS matrix from EMMAX in the combined Pygmy-Bantu sample and analysis-specific Benjamini-Hochberg FDR corrected p-values are given. (XLSX)

**Table S6** Signals of selection from iHS analysis in Bantu samples. Position information, iHS score, the number of SNPs on which it is based, and association results for the top 100 markers showing scores in the empirical distribution of autosomal iHS scores in Bantu individuals. Raw p-values for association with height using the IBS matrix from EMMAX in the combined Pygmy-Bantu sample and analysis-specific Benjamini-Hochberg

FDR corrected p-values are given. (XLSX)

Table S7 PANTHER enrichment analysis. Results are given for genes residing 100 kb up- and downstream of signatures of positive selection and association. Significant results for enrichment near SNPs in the 99.9% tails of the empirical distribution for A) high Pygmy LSBL SNPs, B) XP-EHH SNPs in Pygmies, C) iHS SNPs in Pygmies, D) enrichment for genes 100 kb up- and downstream of SNPs showing raw p-values in the genome-wide EMMAX analysis less than 0.001 in the combined Pygmy-Bantu dataset, E) raw p-values less than 0.001 in the Pygmy-only SNP-based association analyses and F) for similar regions surrounding SupportMix bins with associations less than 0.05. The Insulin/ IGF pathway-protein kinase B signaling cascade, as well as several other pathways involved in immunity, glutamate receptor signaling, metabolism, neuro-endocrine signaling, and reproduction appear enriched in admixture association analysis within Pygmies. Analysis-specific bonferroni corrections for multiple tests were calculated by dividing 0.05 by the total number of pathways

actually tested, less those pathways that include only 1 gene. All pathways showing raw p-values for enrichment less than 0.05 are shown and those meeting the appropriate multiple test correction threshold are given in bold.

## (XLSX)

**Table S8** Selection, association and GWAS results. Results from scans of selection and association analysis in the combined Pygmy-Bantu sample at the 112 SNPs identified in recent GWAS studies of height in non-Africans that were also included in our study. Values for  $F_{ST}$ , LSBL, XPEHH, iHS and association with height using the IBS matrix in the combined Pygmy-Bantu sample are given.

## (XLSX)

**Table S9** Genes showing signatures of selection and/or association with height. Summary of results for all genes 100 kb up- and downstream of SNPs showing both significant signals of positive selection and potential association in one or more of our mapping analyses. Significant results are indicated with an 'x'. Identification of a candidate gene for height via GWAS in non-Africans and membership in various pathways defined using STRING are also shown.

(XLSX)

**Table S10** Admixture mapping results. Association results for the 100 most associated bins using high confidence ancestry estimates from SupportMix. rsID and position information correspond to the first SNP in each 50 SNP bin. Scores and raw p-values using the IBS matrix in EMMAX and analysis-specific FDR corrected p-values along with all genes 100 kb up- and downstream of each of the 50 strongest associations are given. (XLSX)

**Table S11** Top association hits with height for top SNPs in scans for Pygmy-Specific signatures of selection. Position information and association results for SNPs showing both Pygmy-Specific autosomal signatures of selection (LSBL, iHS for Pygmy samples only, and positive XP-EHH) and suggestive association with height in A) the Pygmy samples only (p<0.01) and B) the combined

### References

- Migliano AB, Vinicius L, Lahr MM (2007) Life history trade-offs explain the evolution of human pygmics. Proc Natl Acad Sci U S A 104: 20216–20219.
- Perry GH, Dominy NJ (2009) Evolution of the human pygmy phenotype. Trends Ecol Evol 24: 218–225.
- Batini C, Lopes J, Behar DM, Calafell F, Jorde LB, et al. (2011) Insights into the demographic history of African Pygmies from complete mitochondrial genomes. Mol Biol Evol 28: 1099–1110.
- Destro-Bisol G, Coia V, Boschi I, Verginelli F, Caglia A, et al. (2004) The analysis of variation of mtDNA hypervariable region 1 suggests that Eastern and Western Pygmies diverged before the Bantu expansion. Am Nat 163: 212–226.
- Patin E, Laval G, Barreiro LB, Salas A, Semino O, et al. (2009) Inferring the demographic history of African farmers and pygmy hunter-gatherers using a multilocus resequencing data set. PLoS Genet 5: e1000448. doi:10.1371/ journal.pgen.1000448.
- Quintana-Murci L, Quach H, Harmant C, Luca F, Massonnet B, et al. (2008) Maternal traces of deep common ancestry and asymmetric gene flow between Pygmy hunter-gatherers and Bantu-speaking farmers. Proc Natl Acad Sci U S A 105: 1596–1601.
- Verdu P, Austerlitz F, Estoup A, Vitalis R, Georges M, et al. (2009) Origins and genetic diversity of pygmy hunter-gatherers from Western Central Africa. Curr Biol 19: 312–318.
- Philipson D (1975) The Chronology of the Iron Age in Bantu Africa. The Journal of African History 16: 321–342.
- Tishkoff SA, Reed FA, Friedlaender FR, Ehret C, Ranciaro A, et al. (2009) The genetic structure and history of Africans and African Americans. Science 324: 1035–1044.
- Merimee TJ, Zapf J, Hewlett B, Cavalli-Sforza LL (1987) Insulin-like growth factors in pygmies. The role of puberty in determining final stature. N Engl J Med 316: 906–911.

Pygmy-Bantu sample (FDR<0.05). Analysis-specific Benjamini-Hochberg FDR corrected values are also given. Interestingly, in addition to Insulin/IGF signaling, the thyrotropin-releasing hormone receptor and oxytocin receptor mediated signaling pathways appear to be enriched in an analysis of Pygmy samples only.

# (XLSX)

**Table S12** Top SNPs associated with height near genes involved in the *HGH-IGF1-INS* signaling pathway. The 100 most significantly associated SNPs 100 kb up- and downstream of genes with roles in *GH*, *IGF-1* and *INS* signaling defined via STRING (highest confidence, no more than 50 interactors, 200 additional nodes). Gene name, position information, raw p-values from association analysis for height (IBS matrix) in the combined Pygmy-Bantu sample and in the Pygmy sample only, are given along with analysis-specific FDR values.

(XLSX)

**Table S13** Table of average pairwise  $F_{ST}$  values for all six populations. SNPs with  $F_{ST} = 0$  were omitted from the calculation. (XLSX)

## Acknowledgments

We thank Clara Elbers, Joseph LaChance, Joshua Akey, Andrew Clark, Mingyao Li, Timothy Rebbick, and Scott Williams for their critical feedback and advice, as well as Hakon Hakonarson and the Center for Applied Genomics at the Children's Hospital of Philadelphia for assistance in genotyping the samples. The data reported in this paper are tabulated in the Supporting Information and will be made available at the dbGaP website (www.ncbi.nlm.nih.gov/gap).

## **Author Contributions**

Conceived and designed the experiments: JPJ JM SAT. Performed the experiments: WB BF. Analyzed the data: JPJ LBS SS CL LO BF GH JM SAT. Contributed reagents/materials/analysis tools: AF J-MB GH LO JM. Wrote the paper: JPJ SAT.

- Kang SJ, Chiang CW, Palmer CD, Tayo BO, Lettre G, et al. (2010) Genomewide association of anthropometric traits in African- and African-derived populations. Hum Mol Genet 19: 2725–2738.
- Lango Allen H, Estrada K, Lettre G, Berndt SI, Weedon MN, et al. (2010) Hundreds of variants clustered in genomic loci and biological pathways affect human height. Nature 467: 832–838.
- Merimee TJ, Rimoin DL, Cavalli-Sforza LL (1972) Metabolic studies in the African pygmy. J Clin Invest 51: 395–401.
- Rimoin DL, Merimee TJ, Rabinowitz D, Cavalli-Sforza LL, McKusick VA (1969) Peripheral subresponsiveness to human growth hormone in the African pygmies. N Engl J Med 281: 1383–1388.
- Merimee TJ, Zapf J, Froesch ER (1981) Dwarfism in the pygmy. An isolated deficiency of insulin-like growth factor I. N Engl J Med 305: 965–968.
- Jain S, Golde DW, Bailey R, Geffner ME (1998) Insulin-like growth factor-I resistance. Endocr Rev 19: 625–646.
- Hattori Y, Vera JC, Rivas CI, Bersch N, Bailey RC, et al. (1996) Decreased insulin-like growth factor I receptor expression and function in immortalized African Pygmy T cells. J Clin Endocrinol Metab 81: 2257–2263.
- Baumann G, Shaw MA, Merimee TJ (1989) Low levels of high-affinity growth hormone-binding protein in African pygmies. N Engl J Med 320: 1705–1709.
- Davila N, Shea BT, Omoto K, Mercado M, Misawa S, et al. (2002) Growth hormone binding protein, insulin-like growth factor-I and short stature in two pygmy populations from the Philippines. J Pediatr Endocrinol Metab 15: 269–276.
- Bozzola M, Travaglino P, Marziliano N, Meazza C, Pagani S, et al. (2009) The shortness of Pygmies is associated with severe under-expression of the growth hormone receptor. Mol Genet Metab 98: 310–313.
- Becker NS, Verdu P, Froment A, Le Bomin S, Pagezy H, et al. (2011) Indirect evidence for the genetic determination of short stature in African Pygmies. Am J Phys Anthropol.

- Winkler CA, Nelson GW, Smith MW (2010) Admixture mapping comes of age. Annu Rev Genomics Hum Genet 11: 65–89.
- 23. Akey JM (2009) Constructing genomic maps of positive selection in humans: where do we go from here? Genome Res 19: 711–722.
- Weir BS (1996) Genetic data analysis II : methods for discrete population genetic data. Sunderland, Mass.: Sinauer Associates. xii p445 p.
- Shriver MD, Kennedy GC, Parra EJ, Lawson HA, Sonpar V, et al. (2004) The genomic distribution of population substructure in four populations using 8,525 autosomal SNPs. Hum Genomics 1: 274–286.
- Innan H, Kim Y (2008) Detecting local adaptation using the joint sampling of polymorphism data in the parental and derived populations. Genetics 179: 1713–1720.
- Li JZ, Absher DM, Tang H, Southwick AM, Casto AM, et al. (2008) Worldwide human relationships inferred from genome-wide patterns of variation. Science 319: 1100–1104.
- Voight BF, Kudaravalli S, Wen X, Pritchard JK (2006) A map of recent positive selection in the human genome. PLoS Biol 4: e72. doi:10.1371/journal.pbio. 0040072.
- Sabeti PC, Varilly P, Fry B, Lohmueller J, Hostetter E, et al. (2007) Genomewide detection and characterization of positive selection in human populations. Nature 449: 913–918.
- Young I,J, Wang Z (2004) The neurobiology of pair bonding. Nat Neurosci 7: 1048–1054.
- Kamath J, Yarbrough GG, Prange AJ, Jr., Winokur A (2009) The thyrotropinreleasing hormone (TRH)-immune system homeostatic hypothesis. Pharmacol Ther 121: 20–28.
- Dormitzer PR, Ellison PT, Bode HH (1989) Anomalously low endemic goiter prevalence among Efe pygmies. Am J Phys Anthropol 78: 527–531.
- Lopez Herraez D, Bauchet M, Tang K, Theunert C, Pugach I, et al. (2009) Genetic variation and recent positive selection in worldwide human populations: evidence from nearly 1 million SNPs. PLoS ONE 4: e7888. doi:10.1371/ journal.pone.0007888.
- Eleftherohorinou H, Wright V, Hoggart C, Hartikainen AL, Jarvelin MR, et al. (2009) Pathway analysis of GWAS provides new insights into genetic susceptibility to 3 inflammatory diseases. PLoS ONE 4: e8068. doi:10.1371/ journal.pone.0008068.
- Pickrell JK, Coop G, Novembre J, Kudaravalli S, Li JZ, et al. (2009) Signals of recent positive selection in a worldwide sample of human populations. Genome Res 19: 826–837.
- La Batide-Alanore A, Tregouet DA, Sass C, Siest G, Visvikis S, et al. (2003) Family study of the relationship between height and cardiovascular risk factors in the STANISLAS cohort. Int J Epidemiol 32: 607–614.
- Moore LG, Shriver M, Bemis L, Hickler B, Wilson M, et al. (2004) Maternal adaptation to high-altitude pregnancy: an experiment of nature–a review. Placenta 25 Suppl A: S60–71.
- Moorjani P, Patterson N, Hirschhorn JN, Keinan A, Hao L, et al. (2011) The history of African gene flow into Southern Europeans, Levantines, and Jews. PLoS Genet 7: e1001373. doi:10.1371/journal.pgen.1001373.
- Pool JE, Nielsen R (2009) Inference of historical changes in migration rate from the lengths of migrant tracts. Genetics 181: 711–719.
- Pritchard JK, Di Rienzo A (2010) Adaptation not by sweeps alone. Nat Rev Genet 11: 665–667.
- Kimura R, Fujimoto A, Tokunaga K, Ohashi J (2007) A practical genome scan for population-specific strong selective sweeps that have reached fixation. PLoS ONE 2: e286. doi:10.1371/journal.pone.0000286.
- Tang K, Thornton KR, Stoneking M (2007) A new approach for using genome scans to detect recent positive selection in the human genome. PLoS Biol 5: e171. doi:10.1371/journal.pbio.0050171.
- Williamson SH, Hubisz MJ, Clark AG, Payseur BA, Bustamante CD, et al. (2007) Localizing recent adaptive evolution in the human genome. PLoS Genet 3: e90. doi:10.1371/journal.pgen.0030090.
- Lanktree MB, Guo Y, Murtaza M, Glessner JT, Bailey SD, et al. (2011) Metaanalysis of Dense Genecentric Association Studies Reveals Common and Uncommon Variants Associated with Height. Am J Hum Genet 88: 6–18.
- 45. Ludwig S, Engel K, Hoffmeyer A, Sithanandam G, Neufeld B, et al. (1996) 3pK, a novel mitogen-activated protein (MAP) kinase-activated protein kinase, is targeted by three MAP kinase pathways. Mol Cell Biol 16: 6687–6697.
- Alexander WS, Hilton DJ (2004) The role of suppressors of cytokine signaling (SOCS) proteins in regulation of the immune response. Annu Rev Immunol 22: 503–529.
- 47. de Silva MG, Elliott K, Dahl HH, Fitzpatrick E, Wilcox S, et al. (2003) Disruption of a novel member of a sodium/hydrogen exchanger family and DOCK3 is associated with an attention deficit hyperactivity disorder-like phenotype. J Med Genet 40: 733–740.
- Khor CC, Vannberg FO, Chapman SJ, Guo H, Wong SH, et al. (2010) CISH and susceptibility to infectious diseases. N Engl J Med 362: 2092–2101.
- Yasukawa H, Sasaki A, Yoshimura A (2000) Negative regulation of cytokine signaling pathways. Annu Rev Immunol 18: 143–164.
- Chen Y, Lin G, Huo JS, Barney D, Wang Z, et al. (2009) Computational and functional analysis of growth hormone (GH)-regulated genes identifies the transcriptional repressor B-cell lymphoma 6 (Bc16) as a participant in GHregulated transcription. Endocrinology 150: 3645–3654.

- Kidd JM, Cooper GM, Donahue WF, Hayden HS, Sampas N, et al. (2008) Mapping and sequencing of structural variation from eight human genomes. Nature 453: 56–64.
- Chen J, Billings SE, Nishimune H (2011) Calcium channels link the musclederived synapse organizer laminin beta2 to Bassoon and CAST/Erc2 to organize presynaptic active zones. J Neurosci 31: 512–525.
  Joron M, Frezal L, Jones RT, Chamberlain NL, Lee SF, et al. (2011)
- Joron M, Frezal L, Jones RT, Chamberlain NL, Lee SF, et al. (2011) Chromosomal rearrangements maintain a polymorphic supergene controlling butterfly mimicry. Nature 477: 203–206.
- Redelman D, Welniak LA, Taub D, Murphy WJ (2008) Neuroendocrine hormones such as growth hormone and prolactin are integral members of the immunological cytokine network. Cell Immunol 252: 111–121.
- Ballard C (2006) Strange Alliance: Pygmies in the Colonial Imaginary. World Archaeology 38: 133–151.
- Leonhardt A (2006) Baka and the Magic of the State: Between Autochthony and Citizenship. African Studies Review 49: 69–94.
- 57. Pelican  $\dot{M}$  (2009) Complexities of Indigeneity and Autochthony: An African Example. American Ethnologist 36: 52–65.
- Miller SA, Dykes DD, Polesky HF (1988) A simple salting out procedure for extracting DNA from human nucleated cells. Nucleic Acids Res 16: 1215.
- Purcell S, Neale B, Todd-Brown K, Thomas L, Ferreira MA, et al. (2007) PLINK: a tool set for whole-genome association and population-based linkage analyses. Am J Hum Genet 81: 559–575.
- Team RDC (2007) R: A language and environment for statistical computing. Vienna: R Foundation for Statistical Computing.
  Pemberton TJ, Wang C, Li JZ, Rosenberg NA (2010) Inference of unexpected
- Pemberton TJ, Wang C, Li JZ, Rosenberg NA (2010) Inference of unexpected genetic relatedness among individuals in HapMap Phase III. Am J Hum Genet 87: 457–464.
- Huff CD, Harpending HC, Rogers AR (2010) Detecting positive selection from genome scans of linkage disequilibrium. BMC Genomics 11: 8.
- Scheet P, Stephens M (2006) A fast and flexible statistical model for large-scale population genotype data: applications to inferring missing genotypes and haplotypic phase. Am J Hum Genet 78: 629–644.
- McVean GA, Myers SR, Hunt S, Deloukas P, Bentley DR, et al. (2004) The fine-scale structure of recombination rate variation in the human genome. Science 304: 581–584.
- 65. (2005) A haplotype map of the human genome. Nature 437: 1299–1320.
- Sherry ST, Ward MH, Kholodov M, Baker J, Phan L, et al. (2001) dbSNP: the NCBI database of genetic variation. Nucleic Acids Res 29: 308–311.
- Fujita PA, Rhead B, Zweig AS, Hinrichs AS, Karolchik D, et al. (2011) The UCSC Genome Browser database: update 2011. Nucleic Acids Res 39: D876–882.
- Chakraborty R, Weiss KM (1988) Admixture as a tool for finding linked genes and detecting that difference from allelic association between loci. Proc Natl Acad Sci U S A 85: 9119–9123.
- Sankararaman S, Sridhar S, Kimmel G, Halperin E (2008) Estimating local ancestry in admixed populations. Am J Hum Genet 82: 290–303.
- Pasaniuc B, Sankararaman S, Kimmel G, Halperin E (2009) Inference of locusspecific ancestry in closely related populations. Bioinformatics 25: i213–221.
- Price AL, Tandon A, Patterson N, Barnes KC, Rafaels N, et al. (2009) Sensitive detection of chromosomal segments of distinct ancestry in admixed populations. PLoS Genet 5: e1000519. doi:10.1371/journal.pgen.1000519.
- Bryc K, Auton A, Nelson MR, Oksenberg JR, Hauser SL, et al. (2010) Genomewide patterns of population structure and admixture in West Africans and African Americans. Proc Natl Acad Sci U S A 107: 786–791.
- Tang H, Siegmund DO, Johnson NA, Romieu I, London SJ (2010) Joint testing of genotype and ancestry association in admixed families. Genet Epidemiol 34: 783–791.
- Benjamini Y, Hochberg Y (1995) Controlling the False Discovery Rate: a Practical and Powerful Approach to Multiple Testing. Journal of the Royal Statistical Society Series B (Methodological) 57: 289–300.
- Storey JD, Tibshiranai R (2003) Statistical significance for genomewide studies. PNAS 100: 9440–9445.
- Kang HM, Sul JH, Service SK, Zaitlen NA, Kong SY, et al. (2010) Variance component model to account for sample structure in genome-wide association studies. Nat Genet 42: 348–354.
- Boyko AR, Quignon P, Li L, Schoenebeck JJ, Degenhardt JD, et al. (2010) A simple genetic architecture underlies morphological variation in dogs. PLoS Biol 8: e1000451. doi:10.1371/journal.pbio.1000451.
- Price AL, Zaitlen NA, Reich D, Patterson N (2010) New approaches to population stratification in genome-wide association studies. Nat Rev Genet 11: 459–463.
- Stephens M, Donnelly P (2003) A comparison of bayesian methods for haplotype reconstruction from population genotype data. Am J Hum Genet 73: 1162–1169.
- Stephens M, Scheet P (2005) Accounting for decay of linkage disequilibrium in haplotype inference and missing-data imputation. Am J Hum Genet 76: 449–462.
- Stephens M, Smith NJ, Donnelly P (2001) A new statistical method for haplotype reconstruction from population data. Am J Hum Genet 68: 978–989.
- Bandelt HJ, Forster P, Rohl A (1999) Median-joining networks for inferring intraspecific phylogenies. Mol Biol Evol 16: 37–48.
- Librado P, Rozas J (2009) DnaSP v5: a software for comprehensive analysis of DNA polymorphism data. Bioinformatics 25: 1451–1452.